

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Hannan, Timothy H.; Pilloff, Steven J.

## **Research Report**

Will the Adoption of Basel II Encourage Increased Bank Merger Activity? Evidence from the United States

SUERF Studies, No. 2005/1

### **Provided in Cooperation with:**

SUERF – The European Money and Finance Forum, Vienna

Suggested Citation: Hannan, Timothy H.; Pilloff, Steven J. (2005): Will the Adoption of Basel II Encourage Increased Bank Merger Activity? Evidence from the United States, SUERF Studies, No. 2005/1, ISBN 978-3-902109-26-2, SUERF - The European Money and Finance Forum, Vienna

This Version is available at: https://hdl.handle.net/10419/163466

#### ${\bf Standard\text{-}Nutzungsbedingungen:}$

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# "WILL THE ADOPTION OF BASEL II ENCOURAGE INCREASED BANK MERGER ACTIVITY? EVIDENCE FROM THE UNITED STATES"

by Timothy H. Hannan and Steven J. Pilloff

# CIP

# WILL THE ADOPTION OF BASEL II ENCOURAGE INCREASED BANK MERGER ACTIVITY? EVIDENCE FROM THE UNITED STATES

by Timothy H. Hannan and Steven J. Pilloff

Vienna: SUERF (SUERF Studies: 2005/1)

ISBN 3-902109-26-2

Keywords: Basel II, Capital, Mergers

JEL Classification Numbers: G21, G28, G32, G34

© 2005 SUERF, Vienna

Copyright reserved. Subject to the exception provided for by law, no part of this publication may be reproduced and/or published in print, by photocopying, on microfilm or in any other way without the written consent of the copyright holder(s); the same applies to whole or partial adaptations. The publisher retains the sole right to collect from third parties fees payable in respect of copying and/or take legal or other action for this purpose.

# "WILL THE ADOPTION OF BASEL II ENCOURAGE INCREASED BANK MERGER ACTIVITY? EVIDENCE FROM THE UNITED STATES"

by Timothy H. Hannan and Steven J. Pilloff\*

# Corresponding author:

Timothy H. Hannan
Senior Economist
Division of Research and Statistics
Board of Governors of the Federal Reserve System
Federal Reserve Board
Washington DC 20551
UNITED STATES OF AMERICA

Tel: +1 202-452-2919 Fax: +1 202-728-5838

E-mail: timothy.h.hannan@frb.org

<sup>\*</sup> Board of Governors of the Federal Reserve System. The views expressed are those of the authors and do not necessarily represent those of the Board of Governors of the Federal Reserve System or its staff. The authors would like to thank Robert Avery, Allen Berger, Nora Barger, Ed Ettin, Jim Follain, Diana Hancock, Erik Heitfield, Beverly Hirtle, Myron Kwast, Robin Prager, Kevin Wilson, and staff of the FDIC, OCC, and OTS for comments and suggestions, Shaista Ahmed, David Kite, and Onka Tenkean for excellent research assistance, and Cecilia Tripp for valuable secretarial help.

#### **Abstract**

This study presents two tests of the hypothesis that adoption of an internal ratings-based approach to determining minimum capital requirements, proposed as part of the Basel II capital accord, would cause adopting banking organizations to increase their acquisition activity. The study employs U.S. data and focuses on the advanced internal ratings-based approach, as proposed for banking organizations in the United States. The first test estimates the relationship between excess regulatory capital and subsequent merger activity, including organization and time fixed effects, while the second test employs a "difference in difference" analysis of the change in merger activity that occurred the last time U.S. regulatory capital standards were changed. Estimated coefficients and observed differences have signs consistent with the hypothesis, but results are either statistically insignificant or imply differences that are small in magnitude.

# TABLE OF CONTENTS

Abstract	3
I. Introduction	9
II. Relevant Literature	15
III. Empirical Tests	17
IV. Samples, Data, and Key Variables	23
V. Results	29
VI. Summary and Conclusion	39
Appendix	41
References	49
SUERF – Société Universitaire Européenne de Recherches Financières	51
SHERE STUDIES	53

#### I. Introduction

The Basel II capital accord is expected to change fundamentally the way that the minimum regulatory capital requirements of many banking organizations throughout the world are determined. One of the key elements of this accord is the anticipated introduction of two internal ratings-based (IRB) approaches to regulatory capital requirements. These differ substantially from the Basel I approach and the proposed standardized approach under Basel II in that a banking organization's internal assessments of key risk considerations serve as primary inputs in the calculation of minimum capital requirements. Because these IRB approaches are based on banks' internal assessments using systems validated by supervisors, they offer the benefit of more risk-sensitive regulatory capital requirements.

Two IRB approaches have been proposed. The foundation IRB (F-IRB) approach allows banks to use internal assessments for some risk components and requires them to use regulatory prescribed values for others. The advanced IRB (A-IRB) approach permits banks to use more of their own internal risk assessments. Both IRB approaches will require banks to employ sophisticated risk-measurement techniques that involve a statistical and quantitative assessment of risk, but the demands and complexity of the A-IRB approach are greater than those of the F-IRB approach.

Under the current proposal for banking organizations in the United States, organizations with total banking (and thrift) assets of at least \$250 billion or at least \$10 billion in on-balance-sheet foreign exposure – about ten large organizations based on current balance sheets – would be required to adopt A-IRB.¹ Other banking organizations may also choose to adopt A-IRB, provided they have developed the necessary infrastructure to measure and manage risk. While any bank may "opt in" if it meets regulatory standards, only a few of the largest U.S. banking organizations initially are expected to have in place the infrastructure required to employ such techniques, implying that the A-IRB approach likely will be used at the outset only by a small group of the largest banking organizations. If banks that do not adopt the A-IRB approach continue to be subject to regulatory capital rules based on Basel I, the result would be a bifurcated system in which the largest banking

<sup>&</sup>lt;sup>1</sup> See Board of Governors of the Federal Reserve System (September 2003).

organizations would be subject to the more risk-sensitive and flexible method of determining minimum regulatory capital requirements. It is anticipated that a number of larger banking organizations will join the initial set relatively soon after the implementation date.

In Europe, the application of Basel II is expected to involve more alternatives, as banks will likely have the option of adopting the F-IRB, A-IRB, or standardized approach.<sup>2</sup> Because F-IRB requires fewer internally generated inputs, the option of using this methodology should make it possible for less complex banks to implement an IRB approach to capital requirements. European banks that opt not to adopt an IRB approach will be able to use the standardized approach, which is simpler and less risk sensitive than either of the IRB approaches.

While adoption of an IRB approach may raise the regulatory capital requirements for some IRB banks, it is likely to result in somewhat lower minimum regulatory capital requirements, on average, for the banking organizations that will avail themselves of these approaches, relative to the minimum regulatory capital requirements applied to the majority of banks that initially will not.<sup>3</sup> Concerns have been raised that this disparity would provide an undue competitive advantage to the initial adopters, many of which will be very large banking organizations. Concerns have also been raised that both the excess regulatory capital that would be created at IRB organizations as a result of reduced capital requirements and the aforementioned competitive advantage associated with those reduced requirements would fuel their acquisitions of non-adopting banking organizations. Such concerns have not been the subject of empirical examination. In this paper, we bring data to bear on the second of these concerns: that banking organizations that adopt IRB will aggressively acquire other banking organizations.4

While this issue is relevant to the implementation of Basel II in Europe and elsewhere throughout the world, we restrict our focus in this paper to potential effects on merger activity of the implementation of Basel II in the United

<sup>&</sup>lt;sup>2</sup> See Commission of the European Communities (2004).

<sup>&</sup>lt;sup>3</sup> See "International Convergence of Capital Measurement and Capital Standards: A Revised Framework," which was issued by the Basel Committee on Banking Supervision, Bank for International Settlements, in June 2004.

<sup>&</sup>lt;sup>4</sup> Berger (2004) examines the potential competitive effects of Basel II in the United States on banks in credit markets for small and medium size enterprises.

States, where a bifurcated system is expected to develop with a relatively few very large banks adopting A-IRB and many, smaller banks not adopting A-IRB. There are two primary consequences of adopting an IRB approach to capital requirements that suggest to some observers that IRB banking organizations would increase acquisition activity. Arguments based on these consequences may be usefully designated as "excess regulatory capital" and "relative capital advantage" arguments.

"Excess regulatory capital" arguments assert that merger activity would increase as a result of the excess regulatory capital that would be created by the lower capital requirements stemming from adoption of IRB, and this could fuel acquisitions for a number of different reasons. For example, in the United States, a bank holding company (BHC) desiring to engage in a certain acquisition may be deterred under current capital requirements, because the merger might cause the combined entity to violate existing capital standards. However, a reduction in regulatory requirements and a consequent increase in excess regulatory capital might encourage the acquisition by significantly reducing the likelihood of the combined BHC failing to meet the new, more lenient capital standards.

Another example of an "excess regulatory capital" argument is that, with an increase in excess regulatory capital, adopters could increase their return on equity (ROE) by increasing the amount of earning assets against which a given amount of capital is held (or reducing capital held against a given amount of earning assets). Increased ROE may in turn raise bank valuation, which could facilitate an increase in acquisition activity.

The focus of "relative capital advantage" arguments is on the difference in the capital standards applied to IRB and other banking organizations, maintaining that lower capital requirements for organizations operating under IRB, relative to those operating under existing standards, would result in increased acquisition activity. Specifically, in the U.S. context, some have alleged that A-IRB BHCs would have an incentive to acquire banks not subject to A-IRB capital standards because target banks would be worth more to A-IRB BHCs than to current owners. Different valuations would exist because A-IRB BHCs are expected to face regulatory capital requirements that would be lower than those of the banking organizations that they might acquire. Consequently, A-IRB BHCs could acquire non-adopters and increase the return on equity associated with the acquired assets by either increasing income-earning assets without adding capital or holding less capital against the newly acquired assets.<sup>5</sup>

Both "excess regulatory capital" and "relative capital advantage" arguments rely on the assumption that current regulatory capital requirements are "binding" in the sense that banking organizations are restricted from doing what they would otherwise do in the absence of current minimum capital regulations. Regulatory capital requirements would not be binding if market-based considerations dictated higher levels of capital than those imposed by regulation, or if, as some have argued, "capital arbitrage" techniques currently employed by some larger banking organizations allow them largely to avoid, with minor costs, the constricting effects of existing minimum regulatory capital requirements. However, much of the extensive discussion of the risk implications of Basel II presumes that, at least during crucial periods, capital requirements can be binding and thus influence bank behavior regarding risk.

Ultimately, the question of whether adoption of the Basel II capital accord will result in a substantial increase in merger activity by banking organizations using an IRB approach must be assessed by examining relevant data. The best approach, were it available, would be to examine the results of previous reductions in regulatory capital requirements that applied to some banking organizations but not to others, and assess whether substantial relative increases in the acquisition activity of those granted the reduction occurred as a result of the change. Since no such reduction in capital requirements has taken place in recent decades in the United States, which is the focus of this study, we must assess the issue by pursuing less definitive, but nonetheless informative, approaches.

Specifically, we conduct two different types of tests on U.S. banking data. The first type of test uses recent data on merger activity and BHC capital ratios to determine if, all else equal, large banking organizations with greater

<sup>&</sup>lt;sup>5</sup> Although other arguments for a positive relationship between IRB status and acquisition activity can be made, we believe that the primary reasons that acquisition activity may be affected by IRB depend on "excess regulatory capital" and the "relative capital advantage." An example of an alternative explanation is that the market values the improved ability to measure and manage risk associated with adopting IRB, thereby raising the valuation of IRB organizations and enabling them to increase their acquisition activity. In addition, the costs and benefits associated with an IRB approach could influence decisions by banks not using an IRB approach to merge with each other. This study does not examine the effect of IRB on mergers of this type.

<sup>&</sup>lt;sup>6</sup> See Jones (2002) for a detailed discussion.

<sup>&</sup>lt;sup>7</sup> See in particular Danielsson, et. al. (2001).

excess regulatory capital exhibit a greater tendency to subsequently acquire other banks. Such a finding would be consistent with the argument that allowing large banking organizations to operate under lower capital requirements (and thereby increase excess regulatory capital) would result in expanded acquisition activity on their parts. This approach, however, is subject to several sources of potential endogeneity bias, only some of which are eliminated by the fixed-effects statistical procedure that we employ.

In part for this reason, we also conduct a test based on observations of what happened the last time that capital standards in the United States changed substantially for banks. It is argued that the advent of "prompt corrective action" (PCA) standards, instituted in the early 1990s, increased capital requirements for banks, a change that was in the opposite direction of the reduction of regulatory minimum capital requirements that is expected to occur, on average, for BHCs that adopt the A-IRB approach. 8 Taking a sample of large BHCs that did not appear to be constrained by the capital requirements in effect before the advent of PCA, and, further, would not have been constrained under the pre-PCA capital standards after the adoption of PCA, we compare the change (from the period before to the period after PCA) in merger activity exhibited by those BHCs that did and did not become capital constrained after more stringent regulatory capital standards became relevant. A finding that BHCs constrained by the advent of PCA standards reduced their merger activity by more (or increased it by less) than those not so constrained would be supportive of the hypothesis that relaxation of regulatory capital requirements (as anticipated, on average, for banking organizations that adopt an IRB approach) would result in greater merger activity by IRB organizations.

Our tests are more relevant to the "excess regulatory capital" arguments for increased merger activity by IRB banking organizations than for "relative capital advantage" arguments. However, as discussed below, a number of studies have been conducted that are not supportive of "relative capital advantage" arguments. The results of this literature almost uniformly reject the hypothesis that acquirers seek to purchase more highly capitalized targets – a finding that is not consistent with the notion that acquirers prefer targets with greater potential for ROE improvement from increased leverage.

<sup>8</sup> Although the capital standards of "prompt corrective action" are relevant for banking institutions, not bank holding companies, the amount of capital held by bank holding companies should be affected by the "prompt corrective action" standards.

The plan of the paper is as follows. Section II discusses the literature relevant to the relationship between bank merger activity and capitalization. Section III describes the proposed empirical tests, section IV describes the samples, data, and variables, and section V presents empirical results. A final section summarizes and concludes. To preview results, we do not find convincing evidence that past levels of excess regulatory capital or past changes in capital requirements have had a substantial effect on merger activity. Results of the two tests suggest relationships that are in the direction consistent with the concern that a reduction in minimum capital requirements for banking organizations that adopt IRB would result in increased merger activity on their part, but, with a few exceptions, results are not statistically significant. When results are statistically significant, relevant magnitudes are found to be quite small.

#### II. Relevant Literature

A very large literature has addressed the question of why banking organizations acquire other banking institutions. Several reasons that banks merge have emerged from this literature, and these same reasons are also commonly cited by bankers and other industry analysts. Decifically, the prospects for increased efficiency and gains from diversification are frequently cited as key determinants of acquisition activity, and studies in both Europe and the United States have investigated whether these benefits are in fact realized as a result of mergers. Interestingly, capital is rarely cited as an important issue in the question of why banks merge. Indeed, few studies have sought to investigate the role of capital, especially that of the acquirer's capitalization relative to regulatory requirements. The scarcity of such studies likely reflects the belief that such considerations play a minor role at best in explaining mergers in the banking industry.

The only study that we know of to investigate the acquiring institution's capitalization as a determinant of merger activity was conducted by O'Keefe (1996), who found that acquirers in the large sample of banks that he investigated had significantly lower equity capitalization rates than their nonacquiring peers. Because it suggests that better capitalized banks are less likely to acquire other banks, this finding does not support "excess regulatory capital" arguments that banking organizations, holding increased excess regulatory capital as a result of reductions in minimum regulatory capital requirements, would increase the rate at which they acquire other banking organizations. O'Keefe's sample, however, is not restricted to the very large bank holding companies of concern in this study, so this finding may have limited relevance to the behavior of the BHCs that adopt an IRB approach.

Several studies report evidence relevant to "relative capital advantage" arguments, which our empirical tests do not address very directly. These arguments, as noted above, assert that, with lower capital requirements than

<sup>&</sup>lt;sup>9</sup> For comprehensive reviews, see Berger, Demsetz, and Strahan (1999), Dermine (1999), and Group of Ten (2001), available at <a href="www.bis.org">www.bis.org</a>.

<sup>&</sup>lt;sup>10</sup> Of particular importance to the U.S. banking industry, relaxation of longstanding interstate banking restrictions is widely believed to have sparked extensive consolidation of an industry that was decentralized for over one hundred years.

<sup>&</sup>lt;sup>11</sup> See, for example, Altunbas, Molyneux, and Thornton (1996), Vander Vennet (1996), Akhavein, Berger, and Humphrey (1997), and Amel et. al. (2004).

those of their potential targets, banking organizations employing an IRB approach would have an incentive to acquire these better capitalized targets and increase the return on equity associated with target assets by reducing the capital held against those assets. An implication of this argument for past merger behavior is that acquirers should have found more highly capitalized banks relatively more attractive as acquisition targets.

A fairly large number of studies report results that contradict this implication. We know of at least six studies - Hannan and Rhoades (1987), Amel and Rhoades (1989), O'Keefe (1996), Moore (1997), Wheelock and Wilson (2000), and Akhigbe, Madura, and Whyte (2004) – that sought to determine the characteristics of banking organizations that make them more likely to be a target in a future bank acquisition and that also included the bank's capitalization as a potential determinant. Using various time periods and various samples, five of these six studies find that more highly capitalized banks are less likely, not more likely, to be acquired, all else equal. The sixth study, by Akhigbe, Madura, and Whyte (2004), finds that more capitalized banks are more likely to be acquired, but it differs from the other cited studies in that the sample of targets employed is restricted to relatively large, publicly traded banking organizations. Although the reason for the more common finding of an inverse relationship between target capital ratios and likelihood of acquisition is not clear, the finding is clearly not consistent with the "relative capital advantage" argument.

Another study, by Houston, James, and Ryngaert (2001), addresses in a different way the role of capital as a motivation for bank mergers. As a part of their study, the authors obtained from both managers and analysts opinions and, in some cases, estimates of the sources of expected merger-related gains. Of the 41 mergers on which such information could be obtained, capital structure benefits were noted in only five cases. In four of these cases, analysts noted that the merger might enable the combined bank to free up excess capital, a benefit that would be consistent with "relative capital advantage" arguments. However, because capital is cited in such a small share of the acquisitions in their analysis, their findings seem to suggest that capital has not played a major role in explaining why banks make acquisitions.

## **III.** Empirical Tests

Test 1. Our first test requires estimation of the relationship between BHC capitalization and subsequent BHC merger activity in the United States, using data obtained for recent years. The rationale for this test rests on the presumption that some banking organizations in the recent past have, for whatever reason, found themselves in the position of having capital in excess of the level that they would hold because of regulatory capital requirements, while other banking organizations have found themselves with no such excess and thus may have been constrained by regulatory capital requirements.

The level of capital that BHCs feel bound to maintain because of regulatory requirements may include some additional "cushion" above the required regulatory minimums. Such cushions may be maintained for protection against poor performance or other unanticipated events, and the size of this cushion may differ from one BHC to another, depending on the BHC's risk and other factors.<sup>12</sup>

With this in mind, we seek in this test to determine if BHCs that find themselves with excess regulatory capital exhibit a greater subsequent tendency to acquire other banking organizations than do BHCs that are more constrained by regulatory requirements. The finding of a positive relationship between observed capital ratios and merger activity or the finding of a discreet increase in merger activity at some level of capitalization representing a plausible critical level, would be consistent with the predictions of "excess regulatory capital" arguments that relaxation of capital constraints leads to more merger activity.<sup>13</sup>

<sup>&</sup>lt;sup>12</sup> There is ample evidence that most BHCs choose to maintain some kind of cushion or buffer above minimum regulatory requirements and that its size depends on portfolio characteristics and other factors. See, for example, Hancock and Wilcox (2002).

<sup>&</sup>lt;sup>13</sup> An increase in capital could lead to less, rather than more, merger activity if BHCs with low capital ratios engage in greater acquisition activity than better capitalized BHCs. This could occur because weakly capitalized organizations may purchase highly capitalized targets to increase the capitalization of the combined entity, relative to the pre-merger acquirer. If raising capital levels is a motivation for some mergers, then test results will reflect these mergers, which could obscure the effect of acquisition activity that was conducted for reasons consistent with "excess regulatory capital" arguments. To the extent that such differing types of mergers take place, we believe that results that reflect the average mix of these different types of mergers are the most relevant for understanding the potential effects of IRB.

A point that bears emphasizing, however, is that if the level of capitalization required by the market were greater than that dictated by regulation, or, equivalently, if capital arbitrage allowed BHCs to circumvent regulatory capital requirements with little cost, then there would be little reason to expect a relationship between excess regulatory capital and merger activity.

We can test for this hypothesized relationship by estimating the following relationships:

$$M^{i} = \beta_{0} + \beta_{1}(K/A)^{I} + \beta_{2}X + \varepsilon_{i}, \text{ and}$$
 (1)

$$M^{i} = \alpha_{0} + \alpha_{1}KA1 + \alpha_{2}KA2 + ... + \alpha_{n}KAn + \alpha_{n+1}X + \mu_{i},$$
 (2)

where  $M^i$  denotes the level of merger activity of BHC,  $(K/A)^i$  denotes its capital asset ratio, and KA1, KA2,...KAn denote binary variables that receive values of 1 if  $(K/A)^i$  is in a defined range of values and zero otherwise. Equation (1) imposes a linear relationship between  $M^i$  and  $(K/A)^i$ , while equation (2) allows the relationship to vary across different ranges of  $(K/A)^i$  but not to vary within those ranges. The vector X denotes other explanatory variables that may influence observed merger activity, and  $\varepsilon_i$  and  $u_i$  denote error terms. Finding that  $\beta_i > 0$  in estimations of (1) would be consistent with the hypothesis that merger activity increases with capitalization (and equivalently, excess regulatory capital), and finding that coefficients on KA1, KA2, ..., KAn are positive and increasing in magnitude as capitalization increases would also be consistent with the hypothesis.

If, as noted above, different BHCs set different cushions above the regulatory minimum, estimates of (2) could not be used to identify some critical level of capital below which BHCs are constrained. Under these circumstances, a given binary variable might correctly classify one BHC as not being bound by regulatory requirements, while incorrectly classifying another BHC that was, because of a higher cushion, in fact constrained by such requirements.

Biased estimates attributable to various forms of endogeneity are an important concern in assessing the results of estimations of (1) and (2). Any unobservable characteristic of BHCs that influences both the propensity of a BHC to acquire other banking organizations and its capitalization would impart a bias to the relevant coefficients. To reduce, but unfortunately not eliminate, this possibility, explanatory variables are calculated either for the year prior to that for which merger activity is measured, or, in the case of balance sheet variables, at the beginning of the year for which merger activity

is measured. More importantly, (1) and (2) are estimated using panel datasets consisting of annual observations of large BHCs over two time periods: The first, from 1998 to 2002, is designed to obtain the benefits of panel data estimation using only the most recent (and relevant) five years of available data. A second and longer period, from 1993 to 2002, is also used, since it allows for more annual observations of merger activity and capitalization.

Reported estimations incorporate both year and BHC fixed effects. This approach, in essence, controls for all BHC-specific characteristics that do not vary over time and for all time-specific characteristics that do not vary across BHCs. The inclusion of BHC fixed effects in particular eliminates potential sources of spurious correlation that might arise in comparing one BHC with another.

Spurious correlations in the form of endogeneity bias may result, however, if a time varying unobserved variable influences both a BHC's merger activity and its level of capital (or excess capital) over time in a way different from its effect on other BHCs in the sample. This type of correlation would exist, for example, if BHCs intent on making acquisitions first increase capital levels. The existence of such a correlation between merger activity and measures of capital would bias upward estimates of the coefficients on measures of capital (or excess regulatory capital), resulting in estimates that would overstate the actual expected change in merger activity that would accompany a change in capital requirements.14

Test 2. The second test that we conduct should not be as vulnerable to endogeneity bias but requires that we go back considerably in time to assess the impact on merger activity of a previous change in capital requirements. Specifically, we look at the effect on merger activity attributable to the adoption of more restrictive capital standards in the United States. Passed into law in December 1991 and fully implemented at the end of 1992, the "prompt corrective action" (PCA) provisions of the Federal Deposit Insurance Corporation Improvement Act (FDICIA) of 1991, by all accounts, increased capital adequacy requirements for commercial banks and made more certain that failure to meet them would result in sanctions.

<sup>&</sup>lt;sup>14</sup> Another example might be an improvement in a local economy that resulted in an increase in both the merger activity and excess capital of BHCs located in the relevant area but that had no or less influence on BHCs not located in the area. This would impart a positive bias to the coefficients of the measures of excess capital in the regressions reported below. To control for this possibility, we include a measure of local economic health in the analysis.

In this second test, our strategy, roughly stated, is to examine the change in merger activity exhibited by large BHCs before and after PCA provisions became relevant. We restrict the sample to those BHCs that met the pre-PCA capital requirements both before and after the PCA standards became relevant. These conditions are imposed to determine whether BHCs that became constrained only because of the new capital requirements (and not for other reasons that may entail endogeneity bias) decreased their merger activity by more (or increased it by less) than those BHCs that were not constrained by the new requirements.

A positive answer to the question of whether constrained BHCs decreased their acquisition activity relative to unconstrained BHCs would be consistent with the hypothesis that "binding" or constraining capital requirements reduce merger activity. More relevant to the question at hand, it would be consistent with the hypothesis that relaxation of regulatory capital requirements, to the extent that they are binding or constraining, would result in an increase in merger activity.

A more formal derivation of the test is presented as follows: Suppose that before the advent of PCA, the relationship between the merger activity of a typical, large BHC and its capitalization can be expressed as:

$$M_i^b = \beta_0^b + \beta_1 KADUM^b + \beta_2 X^b + \varepsilon_i^b , \qquad (3)$$

where the superscript "b" denotes that the variable or coefficient pertains to the period before PCA,  $KADUM^b$  is a binary variable that receives the value of one if capitalization was less than the level at which regulatory capital requirements in that period became binding or constraining, and zero otherwise, and  $\varepsilon_i^b$  denotes the error term. The major feature of this specification is that it allows a discrete difference in merger activity for BHCs that do and do not face binding capital constraints.

We posit the same relationship after PCA, expressed as:

$$M_i^a = \beta_0^a + \beta_1 KADUM^a + \beta_2 X^a + \varepsilon_i^a , \qquad (4)$$

where the superscript "a" refers to the period after adoption of PCA standards, with all variables defined as in (3).

Note that coefficients are presumed to be the same in (3) and (4), consistent with the underlying "natural experiment" rationale for the test, which is that

only regulatory capital requirements, and not underlying relationships between acquisition activity and explanatory variables changed between periods. There appears to be little reason to expect changes in regulatory minimums to affect these underlying relationships between merger activity and its determinants. Importantly, KADUM<sup>a</sup> is a binary variable that receives the value of one if capitalization is less than the critical value of capitalization under PCA. If this critical value is higher than that which was relevant in the earlier period, then we will observe some BHCs for which KADUM<sup>b</sup>=0 and KADUM<sup>a</sup>=1, despite little or no change in capitalization. In other words, there will be some BHCs that were not constrained before the introduction of PCA standards, but became constrained as a result of that introduction.

Subtracting (3) from (4) yields:

$$M_i^a - M_i^b = (\beta_0^a - \beta_0^b) + \beta_1(KADUM^a - KADUM^b) + \beta_2(X^a - X^b) + (\varepsilon_i^a - \varepsilon_i^b),$$
 (5)

If determinants of merger activity other than those associated with a binding capital requirement are either invariant over time, (in which case  $X^a = X^b$ ) or, as with variables reflecting the macroeconomic environment, the same across BHCs over the time period, then the term  $\beta_1(X^a - X^b)$  in (5) is either zero or subsumed into the constant term. Under these conditions, only  $\beta_i$ ,  $KADUM^a$ , and KADUM<sup>b</sup> explain differences between the two groups of BHCs in the change in merger activity before and after PCA. It follows that  $(KADUM^a - KADUM^b) = 1$  for the case of a BHC that was not constrained by capital requirements before PCA but was constrained afterwards, and that the term  $(KADUM^a - KADUM^b) = 0$  for BHCs that were not constrained in either period.

If  $\beta_i < 0$ , which is implied if constrained BHCs engage in less merger activity than unconstrained ones, then we have the simple prediction that, of those BHCs believed to be unconstrained by capital requirements prior to PCA  $(KADUM^b = 0)$ , the BHCs that became constrained after the change to tougher capital standards ( $KADUM^a = 1$ ) should have experienced a greater reduction (or smaller increase) in merger activity than those BHCs that remained unconstrained after the change ( $KADUM^a = 0$ ). This prediction follows because the only remaining term (except for the constant) in (5), given these assumptions, is:  $\beta_i(KADUM^a - KADUM^b)$ , and, with  $\beta_i < 0$ , this term is negative for banking organizations constrained by PCA and zero for those that are not.

Under the assumptions discussed above, this prediction may be tested with a straightforward comparison of the change in merger activity across the two groups. This test has the benefit of focusing on the effect of an actual past change in capital standards, and it offers well known advantages associated with this "difference in difference" methodology. Among these advantages, all of the numerous differences in BHCs that might influence merger activity and that do not change over the comparison period "cancel out." Further, because the changes in merger activity for the two groups are calculated for the same time period, the effects of macroeconomic and other changes over time (as long as they influence the two groups equally) are fully controlled for. While these simplifying assumptions appear reasonable, tests based on full estimations of (5) are also conducted.

# IV. Samples, Data, and Key Variables

Samples employed in the analysis consist of the largest U.S. BHCs (based on total assets as of mid-year 2003) that operated throughout the period under investigation. Two successively larger samples of BHCs that operated between year-end 1991 and mid-year 2003 are used in test 1 (panel data analysis). The more restrictive sample includes the ten U.S. banking organizations that are expected to be required to adopt A-IRB status under the current proposals. These organizations are referred to as the mandatory A-IRB BHCs. The first sample also includes the nine other U.S. BHCs with total assets of at least \$50 billion as of mid-year 2003, since they are considered most likely to adopt voluntarily the A-IRB approach in the initial implementation phase. The second and larger sample includes the ten mandatory A-IRB BHCs plus all other U.S. BHCs with total assets of at least \$15 billion as of mid-year 2003. This results in a sample of 38 BHCs and includes a large number of banking organizations that are likely to eventually adopt A-IRB. Analysis is conducted on both samples over two different time periods: a shorter and more recent one covering the years 1998-2002 and a longer one covering the years 1993–2002.

The samples used in test 2 (the natural experiment) are the same as those used in test 1, except that BHCs must have operated between year-end 1986 and mid-year 2003. This requirement causes two BHCs to be dropped because they were not operating during the early part of the period. For reasons discussed below, the years 1987 to 1989 serve as the pre-PCA period, while 1991 and 1992 serve as the post-PCA period.

Merger data were obtained from two sources. The SNL Financial Bank Mergers and Acquisitions Database was the primary source for data on deals that were completed after December 31, 1989. The SNL database includes the vast majority of acquisitions of banks (banks and bank holding companies) and thrifts (savings banks, savings and loan associations, and thrift holding companies) that took place during the period, which includes all of the time covered by test 1 (panel study) and the latter part of test 2 (natural experiment).

The SNL database is not used before 1990 because it is not very comprehensive for deals that took place during that time. Therefore, data for this earlier period were collected from another source. Mergers that took

place in 1987, 1988, or 1989 – the pre-PCA years needed for test 2 – were identified from a database created by staff at the Federal Reserve Board from Federal Reserve Bulletins and reports provided by the Federal Deposit Insurance Corporation (FDIC) and Office of the Comptroller of the Currency (OCC) 15 This database includes only acquisitions in which each party was either a commercial bank or a BHC that operated a commercial bank. Deals involving a thrift as the acquirer or target were not included. Therefore, to maintain consistency, mergers involving savings banks or savings and loan associations that took place after the adoption of PCA are dropped from the group of mergers identified on the SNL database. This requirement has little effect on the set of mergers included in the analysis.

We construct two variables that measure merger activity. The first is the annual number of mergers completed during the relevant period by a BHC. This variable measures the frequency with which a BHC pursued consolidation, not the size of acquisitions. It can be constructed for all years in our analysis and therefore is used for both test 1 and test 2.

A second merger variable that incorporates the size of the targets acquired by a BHC is also used for test 1. Data on the amount of banking assets acquired are available for all years included in test 1 and are used to construct a measure of the relative magnitude of acquisition activity that was conducted by each BHC in the sample. Specifically, the aggregate amount of banking assets acquired in a given year is divided by the BHC's asset level at the start of the year. By dividing by the BHC's total assets, we account for the size of that banking organization.

Capital ratios are constructed with data from the Y-9C report, which is filed quarterly by each BHC with the Federal Reserve Board and contains extensive accounting information on the organization. Creating variables that measure the extent to which BHCs faced capital constraints involves several challenges. First, during the full time period for which data are required (1986-2002), two distinct sets of capital requirements were in effect. Therefore, the capital ratios that are constructed must be those that were relevant at the time.

Capital requirements during the late-1980s predated the implementation of Basel I and its associated risk-based capital rules. At the time, BHCs had to

<sup>&</sup>lt;sup>15</sup> This database was the primary source for two extensive studies of bank merger activity in the United States. See Rhoades (2000) and Rhoades (1996).

satisfy two requirements. First, the ratio of primary capital to assets had to be at least 5.5 percent and the ratio of total capital to assets had to be at least 6 percent. 16 We create variables that measure both the primary and total capital ratios that are consistent with these regulatory requirements.

The original Basel capital accord, or Basel I, was approved in 1988 and fully implemented by 1992. This accord established a new set of capital requirements that attempted to take risk into account. More specifically, less capital had to be held against assets that were considered safer, such as residential mortgages and inter-bank loans, as well as government and agency securities. In the United States, BHCs had to satisfy two risk-based capital requirements and one leverage requirement. The ratio of tier 1 capital to risk-weighted assets had to be at least 4 percent, the ratio of total capital to risk-weighted assets had to be at least 8 percent, and tier 1 capital to average tangible assets had to be at least 4 percent. For each year that the Basel I requirements were in effect, we construct variables that correspond to each of these three capital ratios.

Legal limits represent the lowest level of capital that a BHC can maintain before violating regulatory requirements. However, as mentioned, BHCs are likely to prefer to hold a capital buffer above those regulatory limits for a variety of reasons. First, there are tangible benefits to being considered well or strongly capitalized. During the 1990s, the implementation of PCA standards meant that banks that maintained capital ratios below certain thresholds faced increased regulatory intervention despite the fact that their capital ratios exceeded regulatory minimums. Although PCA does not directly apply to BHCs, it is relevant, because it applies to their bank subsidiaries. To be considered well-capitalized under the requirements of PCA, a bank must have a ratio of tier 1 capital to risk-weighted assets of at least 6 percent, a ratio of total capital to risk-weighted assets of at least 10 percent, and tier 1 capital to average tangible assets of at least 5 percent.

In the late 1980s, prior to Basel I rules, the Federal Reserve Board had established that 7 percent was an important level for the total capital ratio.

<sup>&</sup>lt;sup>16</sup> Primary capital for a bank holding company consists of common stock, perpetual preferred stock, surplus (excluding surplus relating to limited-life perpetual stock), undivided profits, contingency and other capital reserves, mandatory convertible instruments, allowance for possible loan and lease losses (exclusive of allocated transfer risk reserves), minority interest in equity accounts of consolidated subsidiaries, and perpetual debt instruments. Total capital consists of primary and secondary capital. This latter component includes limited-life preferred stock, as well as bank subordinated notes and debentures and unsecured long-term debt of the parent company and its non-bank subsidiaries.

BHCs with total capital that exceeded 7 percent of assets were considered adequately capitalized and faced less intense monitoring and a lower likelihood of supervisory actions than BHCs with ratios below 7 percent, but above 6 percent, the required minimum.

Another reason that BHCs may want to hold capital above Basel I or even the PCA regulatory limits is for protection against downturns in the business cycle and unanticipated events. Additional capital may also be desirable because it would provide BHCs with flexibility that could be used to pursue potentially profitable opportunities such as acquisitions or other types of expansion. Moreover, a buffer may be desirable so that losses do not restrict the BHC's ability to engage in certain businesses. Of course, still another reason that BHCs may maintain capital ratios that exceed regulatory minimums is that the level of capital dictated by the market may exceed the level that would be held because of regulatory requirements.

To make the relationship between excess regulatory capital and merger activity more explicit, measures of BHC capitalization are expressed in terms of excess regulatory capital, which is calculated as actual capital ratios less some critical level based on regulatory requirements or standards. The critical levels chosen for this purpose will be those that must be exceeded to be considered strongly capitalized. More specifically, we use the three ratios required for a bank to be considered-well capitalized under PCA (tier 1 capital to risk-weighted assets of 6 percent, total capital to risk-weighted assets of 10 percent, and tier 1 capital to average tangible assets of 5 percent) for analysis of the years since 1990 and the total capital ratio level (7 percent of assets) required to avoid additional scrutiny in the late 1980s. Although no level for a strong level of the primary capital ratio was defined by the regulator prior to 1991, we use 6.5 percent (the regulatory minimum of 5.5 percent plus 1 percentage point) as an estimate for a primary capital ratio that would be considered a sign of a strong BHC. Table 1 presents a summary of the various capital ratios, requirements, and variables that are relevant for different time periods.

For each BHC, we construct a variable that measures the overall constraint faced by the BHC by taking the minimum of all the measures of excess regulatory capital that were relevant during the year. Because BHCs must satisfy all the capital requirements in effect at a given time, the ratio that reflects the weakest, or most binding, actual capital position is the one that is likely to be most relevant for the BHC. We recognize that simply taking the lowest value is imprecise. For example, ratios are based on different numerators, and, in recent years, different denominators. Nonetheless, we believe that the magnitude of the smallest excess capital measure provides a reasonable proxy for the extent of capital constraints faced by a BHC. For test 1, we measure excess capital at the beginning of each year under investigation (1993-2002 or 1998-2002), and for test 2, we measure it at the beginning of 1987 for the pre-PCA period and at the beginning of 1991 for the post-PCA period.

In short, we take the smallest difference between each of the capital ratios from among the relevant set of regulatory ratios and the value required to be considered strongly capitalized. While not adjusted for individual levels of BHC risk and risk tolerance or for idiosyncratic needs to meet capital requirements, we nonetheless believe that our measure of excess regulatory capital roughly captures the degree to which a BHC faced regulatory capital constraints.

#### V. Results

The two tests that we conduct provide a thorough and rigorous examination of the question of whether changes in regulatory capital requirements might be expected to influence merger activity.

<u>Test 1.</u> In the first type of test conducted, we estimate, for two different samples of BHCs and for two different time periods, the relationship between a BHC's merger activity and its excess capital, defined as the minimum of the difference between each of three actual capital ratio measures and that level of those ratios required to be considered strongly capitalized. Table 2 provides definitions of the independent variables used in these estimations.

Because many of the BHCs in the sample made no acquisitions during at least some of the years in which they are observed, each of the two dependent variables used to measure annual merger activity (the number of acquisitions and the ratio of banking assets acquired to the assets of the acquirer) receive the value of zero for many observations. Because of the well-known violation of OLS assumptions that this entails, other estimation procedures must be used. Since the number of annual acquisitions made by a BHC is a count of the number of occurrences of an event, we use negative binomial maximum-likelihood regression when this variable is employed to measure merger activity. Since the ratio of acquired banking assets to the assets of the acquirer may be thought of as a continuous variable that is censored at zero, we use Tobit maximum-likelihood regressions when this variable is employed to measure merger activity.

Each reported regression includes as an explanatory variable the BHC's expense ratio (*expense ratio*), calculated for the previous year as total noninterest expenses divided by the sum of total noninterest income and net interest income. This rough, but widely used, measure of a BHC's efficiency is included as an explanatory variable because it is often asserted that greater efficiency is associated with greater acquisition activity, as more efficient firms frequently acquire less efficient ones.<sup>18</sup> A negative and significant

<sup>&</sup>lt;sup>17</sup> Because assumptions underlying the more common Poisson maximum-likelihood regressions could be rejected, this more general estimation procedure was chosen. See chapter 19 of Wooldridge (2002) for an extensive discussion on these regression models.

<sup>&</sup>lt;sup>18</sup> Akhavein, Berger, and Humphrey (1997) and Vander Vennet (1996) report evidence consistent with this.

coefficient on this variable would be consistent with this hypothesis, because more efficient firms have lower values of this expense ratio.

To account for differences in the economic conditions in which BHCs in the sample operate, we employ the annual change in housing prices, collected from a weighted repeat sales index (the House Price Index) produced by the Office of Federal Housing Enterprise Oversight. Price changes are measured at the state level and over the same year as merger activity. For BHCs with banking assets in more than one state, a weighted average of these state-specific measures is used, with each state's share of the BHC's total deposits used as the weights.<sup>19</sup>

For each type of estimation, results using two different functional forms are presented, conforming to specifications (1) and (2) above. The first employs as an explanatory variable *excregcap*, which is a continuous measure of excess regulatory capital, measured, as described above, as the minimum difference between each of the three observed capital ratios and that level of each ratio required to be considered well capitalized under PCA. The second replaces this variable with two binary variables indicating different ranges of *excregcap* observed for the BHC. The variable *excregcap*(1–2) indicates that the BHC's excess regulatory capital, measured as described above, is between 1 percentage point and 2 percentage points, while *excregcap*(>2) is similarly defined for BHCs that have excess regulatory capital of at least 2 percentage points. BHC and year combinations in which the minimum capital differential is less than 1 percentage point represent the omitted category.

Tables 3 and 4 report regression results obtained for the 1998–2002 period, and tables 5 and 6 report the results of equivalent estimations conducted for the longer 1993–2002 period. All reported regressions include year fixed effects (reported only in tables 3 and 4 for reasons of space) and BHC fixed effects (not reported in any tables for reasons of space). Tables 3 through 6 each presents four different estimations, organized into two pairs. The first pair reports the results of negative binomial maximum-likelihood regressions when the number of acquisitions serves as the dependent variable, while the second pair present the results of Tobit maximum-likelihood regressions when the ratio of acquired banking assets to assets of the acquirer serves as the dependent variable.

<sup>&</sup>lt;sup>19</sup> We also account for local economic conditions by including a variable that measures the average unemployment rate in the state or states in which a BHC operates. Results (not reported) using this variable are the same as those obtained using the annual change in housing prices.

Consider first table 3, which presents results obtained for the period 1998–2002 for a sample consisting of the ten mandatory A-IRB BHCs and the nine other BHCs with total assets of at least \$50 billion as of mid-year 2003. This sample may be the most immediately relevant to the Basel II proposal, since it is composed specifically of those BHCs whose regulatory capital requirements would be the most likely to be directly affected by the proposal. Of the nineteen BHCs in this group, four are excluded from these estimations because they made no acquisitions during the period.<sup>20</sup>

The finding of positive and statistically significant coefficients on the measures of excess regulatory capital would be consistent with the hypothesis that greater excess regulatory capital enabled or induced BHCs to increase acquisition activity. However, the coefficients reported in 3 are not statistically significant. Indeed the coefficients on *excregcap*, the continuous measure of excess regulatory capital, are negative and insignificant. The coefficients on the binary variable excregcap(1-2) are positive when either the number of mergers or the ratio of assets acquired to total assets is used to measure merger activity, and the coefficient on excregcap(>2) is positive when the ratios of assets acquired to total assets is used, but these coefficients are also not statistically significant, both individually and jointly, with either measure of merger activity.

The coefficients on *expratio* are negative, consistent with the hypothesis that less efficient BHCs (i.e., those with higher expense ratios) are less likely to acquire other banking organizations, but they are not statistically significant. The coefficients on *hpchange*, the change in housing prices during the year, are positive, as might be expected if a better state economy is associated with a greater tendency for BHCs to acquire other banking institutions, but they are also not statistically significant. The coefficients on the year binary variables are negative and, in most cases, statistically significant, reflecting the fact that 1998, the year representing the omitted category, saw a greater amount of merger activity than later years in the period.

Our inability to find a statistically significant relationship between merger activity and excess regulatory capital may reflect the possibility that the level of capitalization required by the market is, for the most part, greater than that required by regulation, with no relationship between regulatory requirements and merger activity the result. However, this lack of statistical significance

<sup>&</sup>lt;sup>20</sup> These observations are dropped, because the fixed-effects statistical model that is used in the empirical analysis requires that, for a given BHC, acquisition activity exhibit some variation over time.

may also reflect the small size of the sample, chosen to contain only those BHCs that are the most likely to be required to adopt or the most likely to adopt voluntarily the A-IRB approach.

Table 4 reports the results of the same regressions, run on a larger sample obtained by lowering the size threshold from \$50 billion to \$15 billion in total consolidated assets as of mid-year 2003. The result is an increase in the number of BHCs in the analysis from 15 to 33 and an increase in the number of year-BHC observations from 75 to 165. For this larger sample, the coefficients on all measures of excess regulatory capital are positive, consistent with the hypothesis that excess regulatory capital induces or enables BHCs to engage in more acquisition activity, but again, none are statistically significant. The estimated coefficients on excregcap(1-2) and excregcap (>2) are jointly insignificant as well.

The coefficients on *expense ratio* are negative, consistent with the hypothesis that less efficient banking organizations exhibit less of a tendency to acquire other organizations, and these coefficients are highly significant when the ratio of assets acquired to total assets is used as the measure of merger activity. The coefficients on *hpchange* are not statistically significant, while the coefficients of the year binary variables are all negative and, in most cases, highly significant, reflecting the general decline in merger activity occurring after 1998.

Tables 5 and 6 report estimates for regressions equivalent to those reported in tables 3 and 4, except that they employ panel data sets that extend from 1993 to 2002 instead of from 1998 to 2002. Table 5 reports these results for a sample consisting only of BHCs with greater than \$50 billion in consolidated assets (which includes all mandatory A-IRB BHCs). Again, all coefficients on variables that measure the degree of excess regulatory capital are positive, but not statistically significant. The coefficients on the two binary excess capital variables are also jointly insignificant.

The coefficients on *expratio* are all negative and, for this sample, highly significant in every case, consistent with the hypothesis that more efficient banking organizations exhibit a greater tendency to acquire other banking organizations. The coefficients on *hpchange* are not statistically significant.

<sup>&</sup>lt;sup>21</sup> Note that the samples analyzed over the 1993-2002 period contain more BHCs than the samples analyzed over the shorter 1998-2002 period, because, for this longer panel, fewer BHCs were omitted as a result of making no acquisitions during the period. Only two BHCs made no acquisitions between 1993 and 2002.

Year fixed effects in the case of this longer panel are not shown for reasons of space.

Table 6 reports the results of equivalent regressions when the sample is expanded to include BHCs with greater than \$15 billion in consolidated assets as of mid-year 2003.

Because of the many years and BHCs included as observations, this sample is the largest of those for which results are reported, and here, we do find statistically significant positive coefficients on the measures of excess regulatory capital when the number of mergers is the measure of merger activity, but not when the ratio of acquired assets to total assets is used as the measure. In this latter case, the coefficients on the binary capital variables are not jointly significant either. Also in the case of this sample, the coefficients on *expense ratio* are negative and statistically significant in most cases, while the coefficients on *hpchange* are not significant.

Summarizing the results reported in tables 3 through 6, we find that coefficients on measures of excess regulatory capital are generally positive, consistent with the hypothesis that excess regulatory capital induces or enables BHCs to increase their level of merger activity, but in most cases are statistically insignificant. Indeed, such coefficients are statistically significant only when the largest sample is employed, and then only for the case in which merger activity is measured by the number of annual acquisitions.

Despite these generally weak regression results, it is still possible that the relationship between excess regulatory capital and BHC merger activity is quantitatively important, based on the magnitude of estimated coefficients. To address this issue, we estimate the likely range of the quantitative impact of adoption of the A-IRB approach on merger activity by combining coefficient estimates with estimates of changes in excess regulatory capital that might result from adoption of A-IRB.

The Third Quantitative Impact Study (QIS 3) was conducted by the Basel Committee on Banking Supervision to understand the possible effects that the Basel II proposals (as of late 2002) might have on capital levels across participating banks.<sup>22</sup> Based on data for 22 U.S. BHCs, the QIS 3 estimated that adoption of A-IRB would, by reducing certain risk weights, lead to an

<sup>&</sup>lt;sup>22</sup> The evolution of the Basel proposal, of course, implies that the QIS 3 may not be a good indicator of the effect of the present proposal. Another quantitative impact study is underway and is expected to be completed in 2005.

average reduction in total risk-weighted assets (RWA) of 6 percent. This change would have the effect of raising the ratios of tier 1 to RWA and total capital to RWA. A change in RWA has no effect on the leverage ratio (total capital to average tangible assets), because the denominator is not based on RWA.

We calculate the three relevant regulatory capital ratios – tier 1 capital to RWA, total capital to RWA, and tier 1 capital to average tangible assets – for each of the 38 BHCs in the sample using data from June 30, 2003. We also estimate the value of those ratios under the A-IRB approach by assuming that RWA would be 6 percent lower than the level reported as of that date. Then, for both sets of the three ratios, we compute the difference between each of the ratios and the minimum needed to be considered well capitalized under PCA standards (see table 1). Next, for both the standard and A-IRB approach, we take the minimum of the three differences. Finally, we subtract the excess capital figure computed under current capital rules from the excess capital figure obtained under the A-IRB approach to get an estimate of the change in excess regulatory capital (expressed as a ratio) that a BHC would experience with the adoption of the A-IRB approach. It should be noted that this final figure will be 0 if the BHC were constrained by the leverage ratio under both capital approaches, since the leverage ratio would be unaffected by adoption of A-IRB.

On average, we find (using QIS 3 results) that adoption of A-IRB would result in an increase in a BHC's excess regulatory capital (expressed as a ratio) of 0.31 percentage points. However, it should be noted that this estimate assumes that every BHC in the sample experienced an identical change in risk-weighted assets equal to the average. In actuality, however, the change in risk-weighted assets following adoption of A-IRB should vary across BHCs. Although not accounting for this variation should affect our estimates of the change in excess regulatory capital, the results of the exercise should not be substantially influenced, because we are estimating the likely range of changes in merger activity, which is rather general.

In order to assess the economic meaning of an increase in excess regulatory capital of 0.31 percentage points, we employ the regression coefficient on *excregcap*, as well as previous levels of BHC merger activity.<sup>23</sup> Calculations of the range of likely changes in acquisition activity are based on the smallest

<sup>&</sup>lt;sup>23</sup> We do not examine the change in merger activity implied by coefficients on *excregcap(1-2)* and *excregcap(>2)* because we estimate that the values of these binary variables following adoption of A-IRB would change for very few BHCs in our sample.

and largest estimated coefficients on *excregcap*, because they generate the most extreme changes in merger activity that can be predicted from regression results.

The largest coefficient estimated for the number of deals is 0.19 (table 4), and it implies that the average number of mergers conducted by a BHC would increase by 6.1 percent, given an increase in excess regulatory capital of 0.31 percentage points.<sup>24</sup> The smallest coefficient estimate is – 0.087 (table 3) and it corresponds to a decrease in the number of acquisitions of 2.7 percent. These percent increases translate into very modest projected changes in merger activity. The average BHC in the full sample of 38 banks conducted 1.74 deals per year between 1993 and 2002, which is greater than the 1998–2002 average for the full sample or any of the averages for the smaller sample of very large BHCs. Given this average number of deals, a 6.1 percent increase would mean an increase in the average annual number of mergers of only 0.1 acquisitions per large BHC, and a 2.7 percent decrease would mean 0.05 fewer acquisitions per year for each large BHC.

With respect to the ratio of acquired banking assets to total BHC assets, the largest coefficient is 0.049 (table 4) and the smallest one is -0.022 (table 3). Respectively, these estimates imply changes in the average value of acquired assets to acquirer assets of 1.5 percentage points and -0.7 percentage points following a 0.31 percentage point jump in excess regulatory capital.<sup>25</sup>

Several caveats suggest that these estimated changes in BHC merger activity that would follow adoption of the A-IRB approach should be viewed as rough, back-of-the-envelope calculations. First, the analysis is static and does not take into account the effect of portfolio changes that could accompany adoption. If BHCs increase the relative share of their assets held in categories that would receive lower risk weights, then the increase in excess regulatory capital could be greater than the estimate of 0.31 percentage points. Second, the data used to estimate changes in RWA are based on QIS 3, which analyzed the effect of the Basel II proposal that was current at the time of the study (late 2002/early 2003). The regulatory capital rules that are ultimately adopted are likely to differ from those used in QIS 3. Finally, we have noted that the estimated change in excess regulatory capital incorporated in this analysis is

<sup>&</sup>lt;sup>24</sup> For the negative binomial regression, the percentage increase in the number of mergers for a given change in excess capital ( $\Delta$ excregcap) can be computed as ( $100 \times e^{(\beta)(\Delta$ excregcap)} - 100), which in this case equals  $100 \times e^{(0.19)(0.31)} - 100$ .

<sup>&</sup>lt;sup>25</sup> The percentage point increase in the ratio of acquired assets to acquirer assets for a given change in excess capital (Δexcregcap) can be computed simply as  $100 \times \beta \times \Delta$ excregcap.

based on the average change in risk-weighted assets and does not take into account the wide range of possible changes that individual BHCs may experience.

In summary, our estimates suggest that the likely change in the number of acquisitions in the United States that would follow adoption of the A-IRB capital approach would fall within a narrow range, and that the number of acquisitions would be unlikely to change much following adoption. This result is especially notable because the only significant results obtained in test 1 are for the case in which acquisition activity is measured by the number of deals. Estimates of the change in the ratio of acquired assets to BHC assets includes more extreme values and the likely range is therefore larger. However, all of these estimates are based on statistically insignificant coefficients.

Test 2. Tables 7 and 8 present the results of t-tests that analyze the effect on merger activity of generally tighter capital requirements brought about by the adoption of PCA capital standards. In these two tables, merger activity before PCA is measured as the average number of mergers per year during the period 1987–1989. Merger activity after the time that the requirements of PCA should have been foreseen, assumed to be the beginning of 1991, is measured as the average number of mergers per year for the period 1991–1992.<sup>26</sup>

Only BHCs judged to be relatively unconstrained by the capital requirements in effect prior to the advent of PCA (the "old standards") are included in the comparisons. Such BHCs are defined as those that met the requirement for being "strongly capitalized" (primary capital ratio of at least 6.5 percent and total capital ratio of at least 7 percent) as of December 31, 1986, the start of our pre-PCA period. In addition, we require that the primary capital and total capital ratios of sample BHCs as of December 31, 1990, the start of our post-PCA period, also exceed 6.5 percent and 7 percent, respectively. Requiring sample BHCs to be well-capitalized under the "old standards" at the start of both periods increases the likelihood that the change in acquisition activity undertaken by a BHC over the analysis period is affected largely by

<sup>&</sup>lt;sup>26</sup> Although PCA was enacted in December 1991 and fully implemented at the end of 1992, we believe that 1991–92, which took place before PCA became legally binding, is the appropriate period to use as the time that PCA standards first became relevant. The data show that in 1991 sample BHCs with less excess regulatory capital, as measured with the ratios relevant under PCA, increased their excess regulatory capital by more than sample BHCs with greater excess regulatory capital measured with those ratios. This behavior suggests that the standards that would become legally effective at the end of 1992 were already affecting BHCs in 1991. Nonetheless, the results of the analysis are similar if 1992–93 is used as the post-PCA period.

the new PCA standards, and is not heavily influenced by any underlying weakness in the BHC's capital position that would have affected merger activity even if standards had not been increased.

BHCs are split into two groups – those that became constrained by the "new standards" introduced by PCA and those that remained unconstrained under these standards. In the post-PCA period, a BHC is classified as constrained if it fails to meet any of the requirements for being well capitalized under PCA (see table 1) as of December 31, 1990. BHCs that meet all three of these requirements are counted as unconstrained after the change to the new capital regime. Clearly, other definitions of what constitutes a binding capital constraint are possible, and we discuss below the results of tests that employ alternative definitions.

Table 7 reports the results of this test for the sample of sixteen mandatory A-IRB BHCs and other BHCs with consolidated assets greater than \$50 billion (as of mid-year 2003) that met the definition of being unconstrained under the "old standards" as of year-end 1986 and year-end 1990. The first row indicates that for those BHCs that became constrained by PCA capital standards (nine BHCs), mergers per year declined from an average of 1.63 during the period 1987–1989 to .61 during the 1991–1992 period. The second row indicates that for the group of BHCs judged not to have become constrained by PCA standards (seven BHCs), mergers actually increased slightly, from 1.33 per year during the 1987–1989 period to 1.50 per year in the period 1991–1992. Perhaps the most relevant number reported in table 7 is the "difference in difference" reported in the third column of the third row. This figure shows that BHCs that became constrained participated in 1.19 fewer mergers per year, on average, than would have been the case had they not become constrained, assuming their merger activity changed after the introduction of PCA in the same way as the group that remained unconstrained. Although the difference between the two groups of BHCs is clearly consistent with the hypothesis that the imposition of binding capital requirements would cause merger activity to decline, the t-statistic calculated for this difference, and reported in the third column of the fourth row, is only – 1.46, indicating that this difference is not statistically significant at levels traditionally employed to reject null hypotheses (in this case that there is no difference between the two groups).

Table 8 reports the results of an equivalent test conducted using a larger sample obtained by including those BHCs that had at least \$15 billion in consolidated assets as of mid-year 2003. In this sample, those BHCs that

became constrained by the requirements of PCA (eleven BHCs) reduced the number of acquisitions that they made annually from 1.48 in the period 1987-1989 to .54 in the 1991-1992 period. The seventeen BHCs in the sample that did not become constrained by the requirements of the new capital standard also exhibited a reduction in average annual acquisitions, from 1.08 in the earlier period to .76 in the later period. The "difference in difference" of -.63 indicates that BHCs that became constrained by PCA participated, on average, in .63 fewer mergers per year than would have been the case had they exhibited the same change as did those BHCs that did not become constrained. The t-statistic of -1.18 registered for this difference indicates that for this sample as well, this difference is not statistically significant.

The results obtained in test 2 appear to be robust, as we conducted a number of alternative analyses and obtained consistent findings. The results of these additional analyses are discussed, but not reported. Results obtained using higher standards to differentiate whether a BHC is classified as capital constrained or not under the new PCA standards indicate no significant differences in the change in merger activity of the two groups of BHCs. In one of these alternative tests, we classified a BHC as constrained if any of its relevant capital ratios were less than the PCA standard plus 0.5 percentage points, and in another alternative, we used the PCA standard plus 1 percentage point to distinguish between constrained and unconstrained BHCs. Results were similar to those reported in tables 7 and 8, which present results based on PCA standards (with no additional cushion) as the level used to classify BHCs.

We also conducted analyses on an expanded sample of 54 BHCs that operated between December 31, 1986 and December 31, 1993 and that held assets of at least \$5 billion as of December 31, 1986. Of these organizations, 45 were unconstrained under the "old standards." Results based on this group also fail to reject the hypothesis of no difference in the change in acquisition activity between those BHCs that became constrained by PCA standards and those that did not. Further, regression analyses that control for other variables that might influence the observed change in a BHC's merger activity, as derived in (5) above, were conducted and found to yield similar, statistically insignificant results. Finally, as noted (see footnote 22), results are similar when the post-PCA period is defined as 1992 and 1993, although more BHCs are unconstrained by PCA standards in this later period.

### VI. Summary and Conclusion

This paper examines empirically the question of whether the internal ratings-based (IRB) approaches employed by certain large banking organizations to determine their regulatory capital requirements, as provided for in the proposed Basel II capital accord, would be likely to lead those banking organizations to increase their acquisition activity. Although the empirical analysis in this study focuses on the United States, where the advanced IRB approach will be implemented by a relatively small number of large banking organizations, the results and discussion are relevant for any country that permits banking organizations to use any internal ratings-based approach to determine capital requirements.

Concerns that acquisition activity would increase following adoption of IRB by a relatively small number of banking organizations stem from two consequences of using an IRB approach. Arguments based on these consequences can be usefully designated as "excess regulatory capital" and "relative capital advantage" arguments. "Excess regulatory capital" arguments focus on the additional excess regulatory capital that would result from a reduction in an IRB adopter's capital requirements as the driver of greater acquisition activity. Arguments based on the "relative capital advantage" cite the disparity in capital requirements that would exist between banking organizations using the IRB approach and those that would not as the force fueling merger activity.

Because we cannot examine the effects of past reductions in capital requirements in the United States, which is the focus of this study, that affect some organizations but not others, we conduct two less definitive, but nonetheless informative, tests. The first uses recent data to determine whether large banking organizations in the United States with greater excess regulatory capital exhibited a greater tendency to subsequently acquire other banking organizations. The second examines whether the generally higher capital requirements resulting from adoption of "prompt corrective action" standards in the United States in the early 1990s (the last time capital requirements were substantially changed) resulted in a relative reduction in merger activity on the part of those large banking organizations most severely affected by the policy.

Both of these tests are most relevant to "excess regulatory capital" arguments for increased acquisition activity. Of relevance to "relative capital advantage" arguments, however, we note that a substantial number of studies do not support a major implication of the arguments.

On the whole, we do not find convincing evidence either that past changes in excess regulatory capital or that past changes in capital standards had substantial effects on merger activity in the United States. Estimated coefficients and observed differences have signs consistent with the concern that a reduction in regulatory capital requirements for large banking organizations would result in increased merger activity on their part. However, results of the two tests are, with a few exceptions, statistically insignificant, and, in cases where results are statistically significant, quantitative magnitudes are small. Overall, we do not find strong evidence that adoption of IRB under Basel II will lead to a substantial increase in merger activity.

# Appendix

Table 1 Capital Measures for U.S. banks

	Relevant years	Regulatory minimum	Minimum for "strongly" capitalized	Source of definition for "strongly" capitalized
Primary capital to total assets*	1987–1989	5.5 %	6.5 %	Same mark-up that is used with total capital
Total capital to total assets*	1987–1989	6.0 %	7.0 %	Top total capital zone established in Federal Reserve System regulations
Tier 1 capital to risk-weighted assets	1990–2002	4.0 %	6.0 %	Prompt corrective action standards
Total capital to risk-weighted assets	1990–2002	8.0 %	10.0 %	Prompt corrective action standards
Tier 1 capital to total assets	1990–2002	4.0 %	5.0 %	Prompt corrective action standards

<sup>\*</sup> The ratios of primary capital to total assets and total capital to total assets are also computed for 1990 in order to identify bank holding companies that were affected by the advent of prompt corrective action standards, but that did not experience a substantial weakening in capitalization under the standards relevant before prompt corrective action.

Table 2

## **Regression Variable Definitions**

Variable	Definition
excregcap	The "excess capital" exhibited by the BHC at the beginning of the year for which merger activity is observed. Since, for the periods examined, specified values of three different types of capital ratios had to be exceeded to be considered well capitalized, this variable measures the smallest of the differences between the observed BHC's capital ratio and that level required to be considered well capitalized under prompt corrective action.
excregcap(1-2)	A binary variable indicating that excregcap is between 1 and 2 percentage points above the regulatory minimum required to be considered well capitalized.
excregcap(>2)	A binary variable indicating that excregcap is 2 or more percentage points above the regulatory minimum to be considered well capitalized.
expense ratio	The "expense ratio" exhibited by the BHC during the year previous to the year for which merger activity is observed, defined as BHC noninterest expenses, divided by the sum of noninterest income and the difference between interest income and interest expenses.
hpchange	Weighted average of the percent change in housing prices in the states in which each BHC operates, with BHC-specific state deposit shares used to calculate the weights.

Table 3

The Relationship Between Merger Activity and "Excess Regulatory Capital" for Mandatory A-IRB BHCs and other U. S. BHCs with Assets over \$50 Billion as of June 30, 2003, Panel Data Estimation for the Period 1998–2002, with Year and BHC Fixed Effects

Dependent variable	Number of Mergers	Number of Mergers	Ratio of Assets Acquired to Total Assets	Ratio of Assets Acquired to Total Assets
excregcap	087 (23)		022 (24)	
excregcap (1–2)		.31 (.80)		.10 (1.03)
excregcap (>2)		.13 (.16)		025 (13)
expense ratio	039 (-1.03)	036 (94)	018 (-1.67)	017 (-1.53)
hpchange	.082 (.58)	.10 (.71)	.017 (.37)	.021 (.46)
Year 1999	44 (98)	33 (69)	35** (-2.68)	35** (-2.70)
Year 2000	87 (-1.52)	75 (-1.31)	48** (-2.68)	47** (-2.79)
Year 2001	-1.05+ (-1.77)	92+ (-1.64)	33+ (-1.93)	33+ (-1.95)
Year 2002	-1.52* (-2.57)	-1.45* (-2.46)	47** (-2.75)	47** (-2.79)
Estimation procedure:	Neg. Binomial regression	Neg. Binomial regression	Tobit regression	Tobit regression
no. of obs.	75	75	75	75
no. of BHCs	15	15	15	15

Note: t-statistics in parentheses. . +, \*, and \*\* indicate significance at the 10, 5, and 1 percent levels, respectively. Four of nineteen BHCs are omitted from the analysis because they made no acquisitions during the time period.

Table 4

The Relationship Between Merger Activity and "Excess Regulatory Capital" for Mandatory A-IRB BHCs and other U. S. BHCs with Assets over \$15 Billion as of June 30, 2003, Panel Data Estimation for the Period 1998–2002, with Year and BHC Fixed Effects

Dependent variable	Number of Mergers	Number of Mergers	Ratio of Assets Acquired to Total Assets	Ratio of Assets Acquired to Total Assets
excregcap	.19 (.99)		.049 (1.09)	
excregcap (1–2)		.31 (1.17)		.069 (.95)
excregcap (>2)		.42 (1.03)		.095 (.89)
expense ratio	026 (-1.13)	031 (-1.32)	016** (-2.68)	017** (-2.75)
hpchange	.089 (1.04)	.10 (1.20)	034 (-1.26)	033 (-1.23)
Year 1999	65* (-2.46)	62* (-2.28)	18* (-2.31)	18* (-2.30)
Year 2000	-1.06** (-2.95)	-1.12** (-3.26)	16 (-1.52)	17 (-1.61)
Year 2001	-1.20** (-3.06)	-1.25** (-3.36)	15 (-1.37)	17 (-1.52)
Year 2002	-1.63** (-4.46)	-1.66** (-4.69)	31** (-3.10)	32** (-3.22)
Estimation procedure:	Neg. Binomial regression	Neg. Binomial regression	Tobit regression	Tobit regression
no. of obs.	165	165	165	165
no. of BHCs	33	33	33	33

Note: t-statistics in parentheses. . +, \*, and \*\* indicate significance at the 10, 5, and 1 percent levels, respectively. Five of thirty-eight BHCs are omitted from the analysis because they made no acquisitions during the time period.

Table 5

The Relationship Between Merger Activity and "Excess Regulatory Capital" for Mandatory A-IRB BHCs and other U. S. BHCs with more than \$50 Billion in Assets as of June 30, 2003, Panel Data Estimation for the Period 1993–2002, with Year and BHC Fixed Effects

Dependent variable	Number of Mergers	Number of Mergers	Ratio of Assets Acquired to Total Assets	Ratio of Assets Acquired to Total Assets
excregcap	.035 (0.24)		.021 (.54)	
excregcap (1–2)		.31 (1.46)		.095 (1.57)
excregcap (>2)		.48 (1.55)		.11 (1.34)
expense ratio	044* (-2.21)	050* (-2.51)	013* (-2.47)	013* (-2.51)
hpchange	067 (-1.15)	050 (-1.40)	0044 (33)	0042 (32)
(Year fixed effects not shown)				
Estimation procedure:	Neg. Binomial regression	Neg. Binomial regression	Tobit regression	Tobit regression
No. of obs.	170	170	170	170
No. of BHCs	17	17	17	17

Note: t-statistics in parentheses. . +, \*, and \*\* indicate significance at the 10, 5, and 1 percent levels, respectively. Two of nineteen BHCs are omitted from the analysis because they made no acquisitions during the time period.

The Relationship Between Merger Activity and "Excess Regulatory Capital" for Mandatory A-IRB BHCs and other U. S. BHCs with more than \$15 Billion in Assets as of June 30, 2003, Panel Data Estimation for the Period 1993–2002, with Year and BHC Fixed Effects

and Dife Fixed Ene	CLS			
Dependent variable	Number of Mergers	Number of Mergers	Ratio of Assets Acquired to Total Assets	Ratio of Assets Acquired to Total Assets
excregcap	.14+ (1.91)		.033 (1.43)	
excregcap (1–2)		.42* (2.32)		.065 (1.38)
excregcap (>2)		.57* (2.49)		.082 (1.39)
expense ratio	018 (-1.52)	021+ (-1.79)	0077* (-2.51)	0080* (-2.62)
hpchange	031 (98)	031 (98)	01 (-1.30)	01 (-1.25)
(Year fixed effects not shown)				
Estimation procedure:	Neg. Binomial regression	Neg. Binomial regression	Tobit regression	Tobit regression
No. of obs.	360	360	360	360
No. of BHCs	36	36	36	36

Note: t-statistics in parentheses. . +, \*, and \*\* indicate significance at the 10, 5, and 1 percent levels, respectively. Two of thirty-eight BHCs are omitted from the analysis because they made no acquisitions during the time period.

Table 7

Yearly Averages of the Number of Mergers Before and After "Prompt Corrective Action" and the Change from Before to After "Prompt Corrective Action" for Mandatory A-IRB BHCs and other U.S. BHCs with greater than \$50 Billion in Assets as of June 30, 2003

45 51 54H 5 50, 2 5 5 5					
	(1)	(2)	(3)		
	Average number of	Average number of	Change		
	mergers per year,	mergers per year,	between the		
	1987–1989	1991–1992	two periods		
(a) BHCs not constrained before but constrained after "prompt corrective					
action" (9 obs)	1.63	.61	-1.02		
(b) BHCs not constrained					
in either period (7 obs)	1.33	1.50	.17		
(c) Difference between					
the groups, (a) – (b)	.30	89	-1.19		
(d) t-statistic for the difference in (c) <sup>1</sup>	.54	-1.35	-1.46		

Assumes unequal variance. Differences may reflect rounding error.

Table 8

Yearly Averages of the Number of Mergers Before and After "Prompt Corrective Action" and the Change from Before to After "Prompt Corrective Action" for Mandatory A-IRB BHCs and other U.S. BHCs with greater than \$15 Billion in Assets as of June 30, 2003

,			
	(1)	(2)	(3)
	Average number of	Average number of	Change
	mergers per year,	mergers per year,	between the
	1987–1989	1991–1992	two periods
(a) BHCs not constrained before but constrained after "prompt corrective			
action" (11 obs)	1.48	.54	94
(b) BHCs not constrained			
in either period (17 obs)	1.08	.76	31
(c) Difference between			
the groups, $(a) - (b)$	.41	22	63
(d) t-statistic for the difference in (c) <sup>1</sup>	.91	64	-1.18

Assumes unequal variance. Differences may reflect rounding error.

### References

- Akhigbe, A., Madura, J., and A. M. Whyte, (2004), "Partial Anticipation and the Gains to Bank Merger," in *Journal of Financial Services Research* 26 (August 2004) pp. 55–71.
- Amel, D. and S. A. Rhoades, (1989), "Empirical Evidence on the Motives for Bank Mergers," in *Eastern Economic Journal* 15 (January–March 1989) pp. 17–27.
- Amel, D., Barnes, C., Panetta, F., and C. Salleo, (2004) "Consolidation and Efficiency in the Financial Sector: A Review of the International Evidence", in *Journal of Banking and Finance* 28 (October 2004) pp. 2493–2519.
- Akhavein, J.D., A.N. Berger, and D.B. Humphrey, (1997) "The Effects of Megamergers on Efficiency and Prices: Evidence from a Bank Profit Function," *Review of Industrial Organization* 12 (February 1997) pp. 95–139.
- Altunbas, Y., Molyneux, P., and J. Thornton, (1997), "Big-Bank Mergers in Europe: An Analysis of the Cost Implications," in *Economica* 64 (1997) pp. 317–29.
- Basel Committee on Banking Supervision, Bank for International Settlements, (2003), *Quantitative Impact Study 3 Overview of Global Results*, (May 5, 2003).
- Basel Committee on Banking Supervision, Bank for International Settlements, (2004), International Convergence of Capital Measurement and Capital Standards: A Revised Framework, (June 2004).
- Berger, A. N., R. S. Demsetz, and P. E. Strahan. "The Consolidation of the Financial Services Industry: Causes, Consequences, and Implications for the Future," *Journal of Banking and Finance* 23 (1999) pp. 135–194.
- Berger, A. N. (2004), *Potential Competitive Effects of Basel II on Banks in SME Credit Markets in the United States*, Board of Governors of the Federal Reserve System, Finance and Economics Discussion Series working paper 2004–12 (March 2004).
- Board of Governors of the Federal Reserve System, (2003), *Basel's Third Quantitative Impact Study: The Results*, (May 2003).
- Board of Governors of the Federal Reserve System, (2003), "Capital Standards for Banks: The Evolving Basel Accord," in *Federal Reserve Bulletin* 89 (September 2003) pp. 395–405.
- Commission of the European Communities, (2004), Proposal for Directives of the European Parliament and of the Council: Re-casting Directive 2000/12/EC of the European Parliament and of the Council of 20, March 2000 relating to the taking up and pursuit of the business of credit institutions and Council Directive 93/6/EEC of 15 March 1993 on the capital adequacy of investment firms and credit institutions, (July 14, 2004).
- Danielsson, J., Embrechts, P., Goodhart, C., Keating, C., Muennich, F., Renault, O., and H. S. Shin, (2001), *An Academic Response to Basel II*, LSE Financial Markets Group, Special Paper No. 130, (May 2001).
- Dermine, J., (1999). *The Economics of Bank Mergers in the European Union, A Review of the Public Policy Issue*, Report commissioned by the Dutch Ministry of Finance
- Group of Ten, (2001), Report on Consolidation in the Financial Sector.

- Hancock, D. and J.A. Wilcox, (2002), Bank Capital and Portfolio Composition, Working Paper, Federal Reserve Board.
- Hannan, T.H. and S.A. Rhoades, (1987) "Acquisition Targets and Motives: The Case of the Banking Industry," in *The Review of Economics and Statistics* 69 (February 1987) pp. 67–74.
- Heckman, J. "Sample Selection Bias as a Specification Error," *Econometrica* 47 (1979) pp. 153–161.
- Houston, J.F., C.M. James, and M.D. Ryngaert. "Where do Merger Gains Come From? Bank Mergers from the Perspective of Insiders and Outsider," *Journal of Financial Economics* 60 (2001) pp. 285–331.
- Jones, D., (2000), "Emerging Problems with the Basel Capital Accord: Regulatory Capital Arbitrage and Related Issues", in *Journal of Banking and Finance* 24 (2000) pp. 35–58.
- Moore, R.R., (1997), *Bank Acquisition Determinants: Implications for Small Business Credit*, Working paper, Federal Reserve Bank of Dallas (April 1997).
- O'Keefe, J.P., (1996), "Banking Industry Consolidation: Financial Attributes of Merging Banks" in *Banking Review* (Fall 1996)
- Rhoades, S.A., (1996), *Bank Mergers and Industrywide Structure*, 1980–94, Board of Governors of the Federal Reserve System, Staff Study 169 (January 1996).
- Rhoades, S. A., (2000), *Bank Mergers and Banking Structure in the United States*, 1980–98, Board of Governors of the Federal Reserve System, Staff Study 174 (August 2000).
- Vander Vennet, R., (1996), "The Effects of Mergers and Acquisitions on the Efficiency and Profitability of EC Credit Institutions", in *Journal of Banking and Finance* 20 (1996) pp. 1531–58
- Wheelock, D.C., and P.W. Wilson, (2000), "Why Do Bank Disappear? The Determinants of U.S. Bank Failures and Acquisitions", in *Review of Economics and Statistics* 82 (February 2000) pp.127–138.
- Wooldridge, J.M., (2002), *Econometric Analysis of Cross Section and Panel Data*, Cambridge: The MIT Press

# **SUERF – Société Universitaire Européenne de Recherches Financières**

SUERF is incorporated in France as a non-profit-making Association. It was founded in 1963 as a European-wide forum with the aim of bringing together professionals from both the practitioner and academic sides of finance who have an interest in the working of financial markets, institutions and systems, and the conduct of monetary and regulatory policy.

SUERF is a network association of central bankers, bankers and other practitioners in the financial sector, and academics with the purpose of analysing and understanding European financial markets, institutions and systems, and the conduct of regulation and monetary policy. It organises regular Colloquia, lectures and seminars and each year publishes several analytical studies in the form of *SUERF Studies*.

SUERF has its full-time permanent Executive Office and Secretariat located at the Austrian National Bank in Vienna. It is financed by annual corporate, personal and academic institution membership fees. Corporate membership currently includes major European financial institutions and Central Banks. SUERF is strongly supported by Central Banks in Europe and its membership comprises most of Europe's Central Banks (29 in total, including the Bank for International Settlements and the European Central Bank), banks, other financial institutions and academics.

### **SUERF STUDIES**

### 1997 - 2002

- 1. **G.M.M. Gelauff and C. den Broeder**, Governance of Stakeholder relationships; The German and Dutch experience, Amsterdam, 1997, ISBN 90-5143-024-8.
- 2. **Rolf Skog**, Does Sweden need a Mandatory Bid Rule, a critical analysis, Amsterdam 1997, ISBN 90-5143-025-6.
- 3. Corporate Governance in Central and Eastern Europe; Transition Management is a Tough Job. Two papers. Amsterdam, 1998, ISBN 90-5143-027-2.
  - 1) **Debora Revoltella**, Financing Firms in East European Countries: An Asymmetric Information and Agency Costs Approach
  - 2) **Peter H. Haiss and Gerhard Fink**, Seven Years of Financial Market Reform in Central Europe
- 4. **Joseph Bisignano**, Towards an Understanding of the Changing Structure of Financial Intermediation; An Evolutionary Theory of Institutional Survival, Amsterdam, 1998, ISBN 90-5143-026-4. *(out of print)*
- 5. **David T. Llewellyn**, The New Economics of Banking, Amsterdam, 1999, ISBN 90-5143-028-0.
- 6. **John Calverley, Sarah Hewin, Kevin Grice**, Emerging Stock Markets after the Crisis, Amsterdam, 2000, ISBN 90-5143-029-9.
- 7. Strengthening Financial Infrastructure: Deposit Insurance and Lending of Last Resort (two contributions), Amsterdam, 2000, ISBN 90-5143-030-2.
  - 1) Richard Dale, Deposit Insurance in Theory and Practice
  - 2) **Christian de Boissieu and Franco Bruni**, Lending of Last Resort and Systemic Stability in the Eurozone.
- 8. **Cem Karacadag and Michael W. Taylor**, The New Capital Adequacy Framework: Institutional Constraints and Incentive Structures, Vienna, 2000, ISBN 3-902109-00-9.
- 9. **Miguel Sebastián and Carmen Hernansanz**, The Spanish Banks' Strategy in Latin America, Vienna, 2000, ISBN 3-902109-01-7.
- 10. **M.M.G. Fase and W.F.V. Vanthoor**, The Federal Reserve System Discussed: A Comparative Analysis, Vienna, 2000, ISBN 3-902109-02-5.
- 11. Willem H. Buiter and Clemens Grafe, Central Banking and the Choice of Currency Regime in Accession Countries, Vienna, 2001, ISBN 3-902-109-03-3.

- 12. **Sanjiva Prasad, Christopher J. Green, and Victor Murinde**, Company Financing, Capital Structure, and Ownership: A Survey, and Implications for Developing Economies, Vienna, 2001, ISBN 3-902109-04-1.
- 13. **Martin M.G. Fase**, Investment in Paintings: The interaction of monetary return and psychic income, Vienna, 2001, ISBN 3-902109-05-X.
- 14. **Alexandre Lamfalussy**, Reflections on the Regulation of European Securities Markets, Vienna 2001, ISBN 3-902109-06-8.
- 15. Italian Mutual Banks: Performance, Efficiency and Mergers and Acquisitions (two contributions), Vienna 2002, ISBN 3-902109-07-6.
  - 1) Juan Sergio Lopez, Alessandra Appennini, Stefania P.S. Rossi, Evidence from two different cost frontier techniques.
  - 2) Roberto Di Salvo, Maria Carmela Mazzilis, Andrea Guidi, Mergers and acquisitions between mutual banks in Italy: an analysis of the effects on performance and productive efficiency.
- Thomas Reininger, Franz Schardax, Martin Summer, Financial System Transition in Central Europe: The First Decade, Vienna, 2002, ISBN 3-902109-08-4.
- 17. **Helmut Wagner**, Implications of Globalization for Monetary Policy, Vienna, 2002, ISBN 3-902109-09-2.
- 18. Luiz Fernando De Paula, Banking Internationalisation and the Expansion Strategies of European Banks to Brazil during the 1990s, Vienna, 2002, ISBN 3-902109-10-6.
- 19. Is there a Future for Regional Banks and Regional Exchanges? The Strategies of Selected Austrian Finance Institutions, (four contributions), Vienna, 2002, ISBN 3-902109-11-4.
  - 1) **David T. Llewellyn**, The Future for Small & Regional Banks in Europe
  - 2) **Reinhard Ortner**, What Future for Regional Banks?
  - 3) Herbert Stepic, The Strategy of RZB in Central and Eastern Europe
  - 4) **Stefan K. Zapotocky**, The Challenges and Chances of Regional Exchanges
- 20. **Thomas Dalsgaard, Jorgen Elmeskov, Cyn-Young Park**, Ongoing Changes in the Business Cycle Evidence and Causes, Vienna, 2002, ISBN 3-902109-12-2.
- 21. **Christian Harm**, Bank Management Between Shareholders and Regulators, Vienna, 2002, ISBN 3-902109-13-0.
- 22. **Jean-Paul Abraham, Peter van Dijcke**, European Financial Cross-Border Consolidation: At the Crossroads in Europe? By Exception, Evolution or Revolution, Vienna, 2002, ISBN 3-902109-14-9.

### 2003

- 2003/1 **Bert Scholtens, Dick van Wensveen**, The Theory of Financial Intermediation: An Essay on What it Does (Not) Explain, Vienna 2003, ISBN 3-902109-15-7.
- 2003/2 **Paola Bongini**, The EU Experience in Financial Services Liberalization: A Model for GATS Negotiation? Vienna 2003, ISBN 3-902109-16-5.
- 2003/3 **Jean-Paul Abraham**, Introduction by **David T. Llewellyn**, Monetary and Financial Thinking in Europe Evidence from Four Decades of SUERF, Vienna 2003, ISBN 3-902109-17-3.
- 2003/4 Securing Financial Stability: Problems and Prospects for New EU Members (three papers) Introduction by **Morten Balling**, Vienna 2003, ISBN 3-902109-18-1
  - 1) **Michael C. Bonello,** Stability Oriented Monetary and Prudential Policies in EU Accession Countries
  - 2) **Fabrizio Saccomanni**, Ensuring Financial Stability: Global and European Perspectives
  - 3) Claudia M. Buch, Jörn Kleinert and Peter Zajc, Foreign Bank Ownership: A Bonus or Threat for Financial Stability?
- 2003/5 **Ralph Süppel**, Russia's Financial Markets Boom, Crisis and Recovery 1995-2001, Vienna 2003, ISBN 3-902109-19-X

### 2004

- 2004/1 Supervisory Systems, Fiscal Soundness and International Capital Movement: More Challenges for new EU Members (three papers) Introduction by **Morten Balling**, Vienna 2004, ISBN 3-902109-20-3
  - 1) **Andreas Grünbichler and Patrick Darlap**, Integration of European Supervisory Systems: Harmonisation or Unification?
  - 2) Sinikka Salo, The Relevance of Fiscal Soundness for Monetary Stability
  - 3) Leslie Lipschitz, Timothy Lane and Alex Mourmouras, How Capital Flowswill influence the EU Accession Countires of Central and Eastern Europe
- 2004/2 European Monetary and Financial Integration: Evolution and Prospects (five speeches), Introduction by **Eduard H Hochreiter** and **David T Llewellyn**, Vienna 2004, ISBN 3-902109-21-1
  - 1) Monetary and Financial Thinking in Europe since the Sixties: Evidence from the SUERF Colloquia. By **Jean-Paul Abraham**

- 2) Fiscal Discipline in a Monetary Union: Issues for the Euro Area. By **Franco Bruni**
- 3) Financial Globalisation and Financial Market Integration in Europe: Challenges Ahead for the European System of Central Banks, by **Baron Alexandre Lamfalussy**
- 4) How to complete the Integration of the European Financial Market, by **Robert Raymond**
- 5) Optimal Currency Areas and Implementation of Economic Policies, by **Jean-Claude Trichet**
- 2004/3 Northern and Eastern Enlargement of EMU: Do Structural Reforms Matter? By Andrew Hughes Hallett, Svend E. Hougaard Jensen and Christian Richter, Vienna 2004, ISBN 3-902109-22-X
- 2004/4 Electronic Purses in Euroland: Why do Penetration and Usage Rates Differ? By **Leo van Hove**, Vienna 2004, ISBN 3-902109-23-8
- 2004/5 From Floating to Monetary Union: The Economic Distance between Exchange Rate Regimes. By **Eduard H. Hochreiter** and **Pierre L. Siklos**, Vienna 2004, ISBN 3-302109-24-6
- 2004/6 Two Measures in Bankruptcy Efficiency. By **Riccardo Brogi** and **Paolo Santella**, Vienna, 2004, ISBN 3-302109-25-4

Order Form: www.suerf.org